Author's response to reviews

Title: When to start antiretroviral therapy in resource-limited settings: a human rights analysis

Authors:

Nathan Ford (nathan.ford@joburg.msf.org)
Alexandra Calmy (alexandra.calmy@hcuge.ch)
Samia Hurst (samia.hurst@unige.ch)

Version: 6 Date: 16 March 2010

Author's response to reviews:

Dear editors

Please find attached a revised manuscript taking into account the latest changes that were requested, as follows:

- Authors' contributions: this has been added
- References 34 and 37: we have deleted the web references as these are no longer functional
- Additional files: we have made the necessary changes. We had originally intended that the tables be placed within the main body of the article but assume that this is not what the editors want. As we do not want to delay publication further we have reluctantly made these changes.

Yours sincerely,

Nathan Ford